ORIGINAL RESEARCH

Evaluating the Therapeutic Efficacy of rTMS Combined with Low-Dose Antipsychotic Medication in Somatic Symptom Disorder

Jinbo Wu¹, Tingting Wang², Xiaomei Jiang¹

¹Department of Psychiatry, The First Affiliated Hospital of Harbin Medical University, Harbin, Heilongjiang, I50001, People's Republic of China; ²Department of Endocrinology, Harbin Red Cross Central Hospital, Harbin, Heilongjiang, 150070, People's Republic of China

Correspondence: Jinbo Wu, Department of Psychiatry, The First Affiliated Hospital of Harbin Medical University, No. 23 Youzheng Street, Harbin, 150001, Heilongjiang, People's Republic of China, Tel +86-0451-85553553, Email tianwenyi526149@163.com

Objective: We aim to evaluate the therapeutic efficacy of repetitive transcranial magnetic stimulation (rTMS) combined with lowdose antipsychotic medication in somatic symptom disorder (SSD) and its effects on neurotransmitters and inflammatory factors.

Methods: This was a retrospective study. According to different treatment regimens, 90 patients with SSD were divided into a medication group (n = 45) and a combination group (n = 45). The medication group received low-dose antipsychotic medication, while the combination group received low-dose antipsychotic medication combined with rTMS. The primary outcomes were to compare the scores of the Hamilton Anxiety Rating Scale (HAMA) and the Hamilton Rating Scale for Depression (HAMD) between the two groups before and after intervention. The secondary outcomes involved assessing the quality of life of the two groups using the Medical Outcomes Study 36-Item Short Form Health Survey (SF-36). Neurotransmitters and inflammatory factors were measured using enzyme-linked immunosorbent assay. Clinical efficacy and adverse reaction rates were also compared.

Results: After treatment, the combination group showed greater improvements in HAMA and HAMD scores, higher SF-36 scores across physical, social, psychological, and daily living dimensions, with increased levels of γ -aminobutyric acid and 5-hydroxytryptamine, decreased dopamine, lower levels of C-reactive protein and interleukin-1 β , higher interleukin-10 levels (all P < 0.05). The total effective rate of the combination group was 97.78%, higher than that of the medication group at 84.44% (P = 0.024). There was no significant difference in the incidence of adverse reactions between the two groups (P > 0.05).

Conclusion: rTMS combined with low-dose antipsychotic medication for SSD shows superiority over medication alone in improving anxiety and depression, enhancing quality of life, regulating neurotransmitter levels, and reducing inflammatory factors, with fewer side effects and significant clinical efficacy. However, due to the small sample size of this study, further prospective, randomized controlled studies with larger samples are needed.

Plain Language Summary: Given the limited efficacy of existing treatments for schizophrenia spectrum disorders (SSD), this study aimed to assess whether combining repetitive transcranial magnetic stimulation (rTMS) with low-dose antipsychotic medication could enhance therapeutic outcomes. By analyzing symptom relief, neuromediator regulation, inflammatory response reduction, and qualityof-life improvements in SSD patients, the results demonstrated that this combined approach outperformed antipsychotic monotherapy. These findings imply that rTMS-antipsychotic combination therapy may represent a viable, innovative treatment option for clinical practice, potentially addressing unmet needs in SSD management.

Keywords: somatic symptom disorder, repetitive transcranial magnetic stimulation, antipsychotic medication, clinical efficacy, neurotransmitters, inflammatory factors

Introduction

Somatic symptom disorder (SSD) refers to a psychiatric condition marked by an intense preoccupation with bodily symptoms.¹ These symptoms cause significant distress or disruption to daily life, evident through exaggerated and

maladaptive cognitive, emotional, and behavioral reactions. For instance, an individual with SSD might exhibit chronic pain, gastrointestinal disturbances, or fatigue, alongside anxiety, catastrophizing thoughts, and avoidance behaviors.² Estimates of the prevalence of SSD in the general population range from 6.7% to 17.4%, with an average frequency of 12.9%.¹ Despite SSD's elevated prevalence and strong association with detrimental functional consequences, such as diminished quality of life and increased healthcare utilization, research on interventions for its treatment and management remains sparse.³ As a result, healthcare providers continue to face challenges in providing adequate treatment for SSD.

Pharmacotherapy is one of the important approaches in the treatment of SSD. Pharmacological treatment for SSD includes non-psychotropic medications (such as β -blockers, non-steroidal anti-inflammatory drugs, and muscle relaxants for alleviating somatic symptoms), psychotropic medications, and herbal remedies (such as St. John's wort). Psychotropic medications have been proven effective in treating some somatic symptoms, including selective serotonin reuptake inhibitors (SSRI) (eg, escitalopram, fluoxetine), serotonin and norepinephrine reuptake inhibitors (SNRI) (eg, venlafaxine), atypical antidepressants (eg, mirtazapine), and tricyclic antidepressants (eg, amitriptyline).⁴ However, a significant number of individuals suffering from SSD do not attain a treatment response, which is characterized by a reduction in severity exceeding 50%, following treatment with antidepressant monotherapy administered at adequate doses and for an adequate duration.⁵ Additionally, antidepressants mainly improve mood and relieve somatic symptoms by regulating neurotransmitters such as 5-hydroxytryptamine (5-HT) and norepinephrine.^{5,6} However, patients with SSD may have more extensive neurotransmitter imbalances, including the dopamine (DA) system.⁷ Antipsychotic medications can regulate DA.⁸ Typical antipsychotic medications include chlorpromazine, trifluoperazine, and pimozide. These drugs, among the earliest developed in the field of antipsychotic medications, primarily exert their effects by targeting DA D2 receptors but may also interact with other receptors.⁹ In the 1990s, new antipsychotic medications are developed, known as second-generation or "atypical" antipsychotic medications, such as quetiapine, aripiprazole, and risperidone.¹⁰ Risperidone, classified as an atypical antipsychotic, exhibits robust antagonistic effects on DA D2. 5-HT2A, 5-HT2C, 5-HT1D, α 1-, and α 2-adrenergic receptors. It serves as an effective augmenting therapy for SSRI-resistant major depressive disorder. Furthermore, risperidone has the capability to counteract the SSRI-induced suppression of norepinephrine activity through its 5-HT2A antagonism.⁵ Paliperidone, which is the principal metabolite of risperidone, has demonstrated notable therapeutic benefits when used as an adjunctive treatment alongside citalopram in patients diagnosed with somatoform disorder.⁵ Compared with other second-generation antipsychotic medications, risperidone has relatively fewer adverse reactions.⁵ However, currently, antipsychotic medications are usually not used as monotherapy for SSD but are part of combination therapy, combined with other treatment methods to improve therapeutic efficacy.⁵

In recent years, the combined application of neuromodulation techniques and pharmacotherapy has provided new ideas for the treatment of SSD.¹¹ Repetitive transcranial magnetic stimulation (rTMS), a form of noninvasive brain stimulation, has the capability to modulate the neural excitability of specific brain regions, which has been applied in the treatment of various neurological and psychiatric disorders.¹² It has few side effects and is considered a promising intervention for improving symptoms in patients with SSD.¹³ rTMS has good efficacy in modulating targeted neural activity and further alleviating symptoms, so an increasing number of studies are focusing on examining the effects of rTMS on negative symptoms and cognitive deficits in patients with SSD.¹³ Xue Li et al report that rTMS combined with family intervention and risperidone has a synergistic effect in the treatment of schizophrenia, especially in improving positive symptoms, negative symptoms, and cognitive function.¹⁴ Additionally, data from Rui Li et al show that rTMS combined with risperidone treatment may affect the brain-gut-microbiota axis by regulating the gut microbiota in patients with chronic schizophrenia, thereby participating in the therapeutic effect.¹¹ Given this, this study aimed to explore the application effect of rTMS combined with low-dose antipsychotic medication (risperidone) in SSD.

It is worth noting that patients with SSD often have comorbid symptoms such as anxiety and depression and may have central nervous system dysfunction.^{2,15} Neurotransmitter imbalance may be one of the mechanisms leading to symptoms and requires further study. It is well known that DA receptors are widely expressed in the body and function in both peripheral and central nervous systems.¹⁶ Their dysfunction is associated with anxiety and obsessive-compulsive symptoms.¹⁷ Furthermore, high levels of γ -aminobutyric acid (GABA) in the medial prefrontal cortex play an important pathophysiological role in the generation of somatic symptom disorders.¹⁸ Meanwhile, 5-HT abnormalities are also considered a key biological cause of somatic symptoms.⁵ On the other hand, data from Bumhee et al showed that patients with SSD have higher levels of interleukin-6 (IL-6) and tumor necrosis factor-alpha (TNF- α) compared with healthy

controls, and hypersensitive C-reactive protein (hs-CRP) and IL-6 have a complete mediating inhibitory effect on the relationship between the functional connectivity strength of the default mode network and depression levels.¹⁹ Against this background, we included studies on neurotransmitters and inflammatory markers to comprehensively evaluate the biological effects of rTMS combined with risperidone on SSD. By analyzing changes in neurotransmitters such as DA, GABA, 5-HT, and inflammatory markers, we can reveal the potential mechanisms of combination therapy and provide a scientific basis for the treatment of SSD.

Materials and Methods

Ethics Statement

All experimental procedures were approved by the Medical Ethics Committee of The First Affiliated Hospital of Harbin Medical University, and patients or their families provided informed consent and signed an informed consent form. This study adhered to the Declaration of Helsinki.

Subjects and Grouping

Ninety patients with SSD admitted to The First Affiliated Hospital of Harbin Medical University from May 2023 to May 2024 were selected as the study subjects. Inclusion criteria: Patients who met the diagnostic criteria for SSD in the International Classification of Diseases, 10th Edition²⁰ and the Chinese Classification and Diagnostic Criteria for Mental Disorders, 3rd Edition, and had a Patient Health Questionnaire-15 score ≥ 5 ; patients aged 18–75 years; patients who had not received antipsychotic medication or rTMS within the past 6 months; patients on the 17-Item Hamilton Rating Scale for Depression (HAMD)²¹ score ≥ 17 points before treatment. Exclusion criteria: Patients with severe physical diseases (including stroke, malignant tumors, chronic obstructive pulmonary disease, end-stage renal disease, liver cirrhosis, myocardial infarction, etc) or brain organic diseases; patients with a history of schizophrenia, mania, and depression associated with various types of organic lesions; patients with epilepsy, acute-phase cerebrovascular accidents, intracranial infections, and intracranial presence of metal or other foreign bodies; patients with suicidal behavior or suicidal ideation; patients with allergies or sensitivities to all the medications used in the study; and women who were pregnant or breastfeeding.

Depending on the treatment regimen, the enrolled patients were divided into the medication and combination groups, with 45 cases each. The medication group was treated with low-dose antipsychotic medication, while the combination group received low-dose antipsychotic medication combined with rTMS.

Treatments

Patients in the medication group were treated with low-dose antipsychotic medication, specifically risperidone (Janssen Pharmaceutical Co., Ltd., Xi 'an, China; State Drug Administration H20010309), administered orally at an initial dose of 0.5 mg twice daily, with the dose adjustable to 1.5 mg within one week, for a treatment cycle of 8 weeks. The combination group received low-dose antipsychotic medication combined with rTMS. Specific methods: The antipsychotic medication regimen was identical to that of the medication group. YRDCCY-1 transcranial magnetic stimulator (Yi Ruide Medical Equipment New Technology Co., Ltd., Wuhan, China) was used for the treatment, with an output pulse frequency range of 0–100 Hz and a peak stimulation intensity range of 1.5–6 Tesla. Specific operations: The patient was laid on the magnetic therapy bed, relaxed systemically, with the "8"-shaped coil placed on the right dorsolateral prefrontal cortex. The treatment was stimulated with an intensity of 1Hz frequency and 80% motor threshold, delivering 800 pulses per session, lasting 20 minutes per session, once daily for 5 consecutive days, followed by a 2-day break. This cycle was continued for 8 weeks.

Observation Parameters

General data: General demographic information such as gender, age, years of education, ethnicity, marital occupation, and place of residence of the study subjects were collected through a retrospective case system.

Anxiety and depression: The Hamilton Anxiety Rating Scale $(HAMA)^{22}$ was used to assess patients' psychological states. The scale includes 14 items, such as anxious mood, fear, and cognitive function, scored on a 0–4 scale (0 = none, 1

= mild, 2 = moderate, 3 = severe, 4 = very severe), with a maximum score of 56. The HAMD-17 scale was used to assess the depressive symptoms in both groups of patients. This scale consists of 17 items covering aspects such as depressed mood, guilt, and suicidal thoughts, and is divided into five factors: anxiety somatization, weight, cognitive disturbance, retardation, and sleep disturbances. Each question has a different scoring standard, and the corresponding score is selected based on the patient's response. The total score is then calculated.

Clinical efficacy: HAMD-17 was utilized to evaluate the treatment response. A reduction rate of \geq 75% before and after treatment was considered clinical remission, 25–74% was considered effective, and < 25% was considered ineffective. The total effective rate was calculated as [(number of clinical remissions + effective cases) / total number of cases × 100%]. The reduction rate was calculated as [(pre-treatment HAMD-17 score - post-treatment HAMD-17 score) / pre-treatment HAMD-17 score × 100%].

Quality of life: The Medical Outcomes Study 36-Item Short Form Health Survey (SF-36) (SF-36)²³ was used for assessment. The survey included four functional domains: physical, social, psychological, and daily living. Each domain is scored out of 100, with higher scores indicating better quality of life for the patient.

Adverse reactions: The occurrence of adverse reactions including fatigue, drowsiness, anorexia, nausea, xerostomia, and dizziness, were compared.

Neurotransmitters levels: Before treatment (admission day 2) and after treatment (8 weeks of treatment), 5 mL of cubital vein blood was collected with an EDTA blood collection tube. Blood samples should be processed within 2 hours; otherwise, they must be stored at -80° C, with strict avoidance of multiple freeze–thaw cycles. The levels of GABA, 5-HT, and DA were measured using the enzyme-linked immunosorbent assay (ELISA).²⁴ ELISA kits for GABA (CB10292-Hu), 5-HT (CB10030-Hu), and DA (CB10524-Hu) were purchased from Coibo Biotechnology (Shanghai, China).

Inflammatory factor levels: Levels of CRP, IL-1 β , and IL-10 were measured using ELISA. ELISA kits for CRP (CB10116-Hu), IL-1 β (CB10347-Hu), and IL-10 (CB13566-Hu) were available from Coibo.

Statistical Processing

Data were processed using SPSS 26.0 software. The normality of measurement data was tested using the Shapiro–Wilk method, and all data conformed to a normal distribution. Additionally, the measurement data were expressed as mean \pm standard deviation (Mean \pm SD). Comparisons between groups used independent sample t-tests, and within-group comparisons used paired sample t-tests. Numeration data were expressed as [number of cases (n)], and analyzed using chi-square tests, with Yates' correction applied to small sample sizes. A *P*-value < 0.05 was considered statistically significant.

Results

General Data

Baseline data differences between the two groups were analyzed, revealing no statistically significant differences in age, gender, disease duration, marital status, ethnicity, residence, or education level (P > 0.05), indicating comparability (Table 1).

Information		Medication Group (n = 45)	Combination Group (n = 45)	χ ² /t	Р
Age Disease duration (years)		40.31 ± 7.83 5.11 ± 2.12	40.96 ± 7.60 4.85 ± 2.23	-0.396 0.561	0.693 0.576
Gender	Male Female	15 30	19 26	0.756	0.384

(Continued)

Information		Medication Group (n = 45)	Combination Group (n = 45)	χ ² /t	Р
Marital status	Unmarried	8	11	1.042	0.594
	Married	36	32		
	Divorce	I	2		
Ethnicity	Han nationality	41	43	0.714	0.398
	Ethnic minority	4	2		
Residence	Rural	18	22	1.131	0.288
	Urban	27	23		
Education level	Primary school	9	12	1.209	0.751
	Middle school	16	19		
	High school	10	7		
	College degree or above	10	7		

Table I (Continued).

HAMA and HAMD Scores

There was no notable difference in HAMA and HAMD scores between the two groups before treatment (P > 0.05). After treatment, HAMA and HAMD scores in both groups decreased compared to pre-treatment levels (P < 0.05). Post-treatment improvements in HAMA and HAMD scores were better in the combination group than in the medication group (Table 2).

SF-36 Scores

There was no significant difference in SF-36 scores between the two groups before treatment (P > 0.05). After treatment, SF-36 dimension scores in both groups increased (P < 0.05), with the combination group's scores higher than the control group's (P < 0.05) (Table 3).

Clinical Efficacy

The total effective rate in the medication group was 84.44% (38/45), while the total effective rate in the combination group was 97.78% (44/45). According to the chi-square test, the total effective rate in the combination group was higher than that in the medication group (P = 0.026) (Table 4).

Occurrence of Adverse Reactions

The medication group and the combination group showed no significant difference in the incidence of adverse reactions such as fatigue, drowsiness, anorexia, xerostomia, nausea, and dizziness (all P > 0.05) (Table 5).

Group	HAMA (Score)		HAMD (Score)		
	Before Treatment	After Treatment	Before Treatment	After Treatment	
Medication group (n = 45)	21.22 ± 5.12	13.22 ± 3.27 ^a	23.73 ± 4.12	12.76 ± 3.58 ^a	
Combination group (n = 45)	20.84 ± 5.26	9.20 ± 2.07^{a}	23.76 ± 3.76	9.09 ± 2.84 ^a	
t	0.345	6.962	-0.027	5.379	
Р	0.731	<0.001	0.979	< 0.001	

Table 2 Comparison of HAMA and HAMD Scores Before and After Treatment in the Two Groups

Note: ^a *P* < 0.05 compared to pre-treatment in the same group.

Group	Physical		Social		
	Before Treatment	After Treatment	Before Treatment	After Treatment	
Medication group (n = 45)	60.24 ± 6.21	73.24 ± 4.55 ^a	52.22 ± 5.19	69.27 ± 6.78 ^a	
Combination group (n = 45)	61.29 ± 6.12	84.29 ± 5.31 ^a	51.60 ± 4.90	$78.27 \pm 6.24 a$	
t	-0.804	-10.595	0.584	-6.553	
Р	0.423	< 0.001	0.56	< 0.001	
	Psychological				
Group	Psychological		Daily living		
Group	Psychological Before treatment	After treatment	Daily living Before treatment	After treatment	
Group Medication group (n = 45)	, ,	After treatment 67.69 ± 3.28 ^a	, ,	After treatment 70.18 ± 4.41 ^a	
	Before treatment		Before treatment		
Medication group (n = 45)	Before treatment 50.13 ± 4.46	67.69 ± 3.28 ^a	Before treatment 56.53 ± 5.32	70.18 ± 4.41 ª	

Table 3 Comparison of SF-36 Scores Before and After Treatment Between the Two Groups

Note: ^a P < 0.05 compared to pre-treatment in the same group.

Table 4 Comparison of Clinical Efficacy Between the Two Groups

Group	Clinical Remission	Effective	Ineffective	Total Effective Rate
Medication group (n = 45)	5	33	7	38 (84.44%)
Combination group (n = 45)	9	35	I	44 (97.78%)
χ^2	-	-	-	4.939
Р	-	-	-	0.026

Table 5 Comparison of the Incidence of Adverse Reactions Between the Two Groups

Group	Fatigue and Drowsiness	Xerostomia	Anorexia	Nausea	Dizziness
Medication group (n = 45)	6	5	4	2	3
Combination group (n = 45)	4	4	2	3	5
χ^2	0.450	0.123	0.714	0.212	0.549
Р	0.502	0.725	0.398	0.645	0.459

Neurotransmitters Levels

There was no significant difference in GABA, 5-HT, and DA levels between the two groups before treatment (P > 0.05). After treatment, GABA and 5-HT levels in both groups increased compared to pre-treatment levels, while DA levels decreased, with the combination group's levels higher/lower than the medication group's (Table 6).

Inflammatory Factor Levels

There was no remarkable difference between the pre-treatment levels of CRP, IL-1 β , and IL-10 between the two groups (P > 0.05). After treatment, CRP and IL-1 β levels in both groups decreased compared to pre-treatment levels, while IL-10 levels increased, with the combination group's levels lower/higher than the medication group's (Table 7).

Discussion

SSD is common in primary health care institutions, and diagnosis and treatment are often challenging.¹ Therefore, this study evaluated the therapeutic effects of rTMS combined with low-dose antipsychotic medication in patients with SSD. The results showed that rTMS combined with low-dose antipsychotic medication has significant advantages in improving anxiety and depression, enhancing quality of life, and regulating neurotransmitter levels, and inflammatory factors in the treatment of SSD.

Group	GABA (ng/L)		5-HT (µg/L)		DA (ng/mL)	
	Before Treatment	After Treatment	Before Treatment	After Treatment	Before Treatment	After Treatment
Medication group (n = 45)	5.72 ± 1.33	9.33 ± 2.34 ª	198.45 ± 30.23	$241.23 \pm {}^{23.57 a}$	16.72 ± 2.01	13.62 ± 1.54 ^a
Combination group (n = 45)	5.62 ± 1.42	11.72 ± 3.14 ª	193.23 ± 29.22	275.56 ± 20.56 ^a	16.34 ± 2.13	10.73 ± 1.26 ^a
t P	0.348 0.729	-4.093 < 0.001	0.833 0.407	−7.363 < 0.001	0.867 0.388	9.743 < 0.001

Table 6 Comparison of Neurotransmitters Levels Between the Two Groups Before and After Treatment

Note: ^a P < 0.05 compared to pre-treatment in the same group.

Table 7 Comparison of Inflammato	ry Factor Levels Between the Two Grou	ups Before and After Treatment
----------------------------------	---------------------------------------	--------------------------------

Group	CRP (mg/L)		IL-Iβ (ng/L)		IL-10 (pg/mL)	
	Before Treatment	After Treatment	Before Treatment	After Treatment	Before Treatment	After Treatment
Medication group (n = 45)	19.33 ± 4.12	13.38 ± 2.23 ª	83.23 ± 9.76	65.42 ± 5.27 ª	7.36 ± 2.21	13.33 ± 3.23 ^a
Combination group (n = 45)	19.52 ± 4.33	10.05 ± 2.07 ^a	82.29 ± 10.06	52.34 ± 4.61 v	7.72 ± 2.10	17.52 ± 2.87 ^a
t P	-0.213 0.831	7.34 < 0.001	0.451 0.653	2.53 < 0.00	-0.794 0.429	-6.504 < 0.001

Note: ^a P < 0.05 compared to pre-treatment in the same group.

Specifically, the improvement in HAMA and HAMD scores in the combination group was better than that in the medication group, indicating that rTMS combined with low-dose antipsychotic medication is more effective in improving anxiety and depressive symptoms. Additionally, the SF-36 scores in the combination group were higher than those in the medication group, indicating that combination therapy is more effective in improving quality of life. Meanwhile, the total effective rate in the combination group was higher than that in the medication group, and there was no significant difference in the incidence of adverse reactions between the two groups, revealing that rTMS combined with low-dose antipsychotic medication is superior to monotherapy in overall efficacy and does not increase the risk of adverse reactions. Studies have reported that rTMS leads to a greater reduction in depressive symptoms than medication, which is further reflected in higher rates of response and remission. Additionally, rTMS results in a more substantial decrease in anxiety and anhedonia symptoms compared to switching antidepressants.²⁵ Furthermore, in the study of Ren et al, intermittent theta burst-rTMS (iTBS-rTMS) demonstrates a favorable therapeutic effect in patients with methamphetamine use disorder, with improvements in both depression and impulsivity. These enhancements are strongly associated with the therapeutic efficacy of iTBS-rTMS.²⁶

In addition, this study highlighted that the levels of GABA and 5-HT in the combination group were higher than those in the medication group, and the level of DA was lower than that in the medication group, indicating that rTMS combined with low-dose antipsychotic medication is more effective in regulating neurotransmitter levels. According to previous results, the serum levels of 5-HT in both the acupuncture + rTMS combination group and the rTMS-only group are found to be higher post-treatment compared to their pre-treatment levels.²⁷ In the study by Feng et al, after rTMS treatment, the GABA levels are notably elevated.²⁸ A study reveals that antipsychotic medications such as risperidone may regulate N-methyl-D-aspartate receptors through the glycogen synthase kinase 3β - β -catenin signaling pathway and/or the activation of cyclic adenosine monophosphate response element-binding protein 1, and the regulation of GABA_aR may also be related to these signaling pathways.²⁹ Thomas et al's study shows that risperidone affects motor activity and neural

activity in mice, especially having a significant impact on neural oscillations in the prefrontal cortex and hippocampus, by blocking DA receptor D₂R and regulating the activity of serotonin receptors (5-HT₁AR and 5-HT₂AR).³⁰ Based on the results of this study, there is a synergistic mechanism in neurotransmitter regulation when risperidone is combined with rTMS. Risperidone, as a D2 receptor antagonist, can block excessive DA activity in the mesolimbic system to alleviate positive symptoms,¹⁴ while rTMS activates glutamatergic neurons in the prefrontal cortex through high-frequency stimulation, promoting DA release and enhancing prefrontal-striatal circuit function. The combined treatment may partially counteract the inhibitory effect of risperidone on DA, thereby optimizing DA metabolic balance. Additionally, risperidone indirectly increases synaptic 5-HT levels by blocking 5-HT2A receptors, and rTMS stimulates the dorsolateral prefrontal cortex to increase 5-HT release into the limbic system. The synergy of the two may increase 5-HT concentrations in relevant brain regions, thereby improving anxiety and depressive symptoms. On the other hand, rTMS can enhance GABAergic interneuron activity to inhibit prefrontal cortex hyperactivity, and risperidone can indirectly regulate DA-GABA interactions to enhance GABAergic inhibition. The combined treatment can more effectively reduce prefrontal cortex excitability and reduce anxiety and obsessive-compulsive symptoms.

Meanwhile, this study also observed that the levels of CRP and IL-1 β in the combination group were lower than those in the medication group, and the level of IL-10 was higher than that in the medication group, indicating that rTMS combined with low-dose antipsychotic medication is more effective in regulating inflammatory marker levels. In line with the results of the present study, in the study of Yang et al, deep rTMS reduced microglial activation at the lesion sites and normalized cytokine levels (IL-1 β , IL-6, and IL-10) in regions affected by cuprizone.³¹ Based on the results of this study, rTMS may activate the vagus nerve-cholinergic anti-inflammatory pathway, inhibit microglial activation, and reduce the release of pro-inflammatory cytokines (such as IL-1 β). Although risperidone does not directly have antiinflammatory effects, it may inhibit microglial M1 polarization by regulating the DA signaling pathway. When combined, the two may synergistically reduce peripheral blood CRP and IL-1 β levels and increase the level of the anti-inflammatory cytokine IL-10.

Limitation

This study preliminarily revealed the efficacy of rTMS combined with risperidone in the treatment of SSD in improving anxiety and depression, enhancing quality of life, regulating neurotransmitter levels, and inflammatory factors. However, this study also has some limitations. For example, we did not perform sample size calculation, and the sample size was small, which may affect the generalizability of the findings. Additionally, the 8-week observation period in this study may not be sufficient to assess the long-term efficacy and potential delayed effects of rTMS combined with risperidone treatment. The lack of long-term follow-up may not allow for the assessment of symptom recurrence rates and drug dependence after treatment.

Conclusion

In conclusion, this study confirms that rTMS combined with low-dose antipsychotic medication is superior to monotherapy in improving anxiety and depression, enhancing the quality of life, regulating neurotransmitter levels, and inflammatory factors in the treatment of somatic symptom disorder, with fewer side effects and significant clinical efficacy. This study provides a new direction for the treatment of SSD, and it is hoped that the treatment regimen can be further optimized to benefit more patients in the future. Importantly, future research should further verify the mechanisms and long-term safety of combination therapy by expanding the sample size, extending the follow-up period, and controlling confounding factors (such as treatment adherence, social support systems, etc). Additionally, the treatment of SSD requires multidisciplinary collaboration, including psychiatry, neurology, and psychology. The multidisciplinary collaborative treatment model can integrate the advantageous resources of various disciplines to provide patients with more personalized and comprehensive treatment plans and improve the overall treatment level of SSD.

Ethical Statement

All experimental procedures were approved by the Medical Ethics Committee of The First Affiliated Hospital of Harbin Medical University, and informed consent was signed by patients' families.

Consent to Participate

Informed consent was obtained from all individual participants included in the study.

Funding

No funds, grants, or other support was received.

Disclosure

The authors declare that they have no conflicts of interest for this work.

References

- 1. Miyoshi M, Takanashi R, Taguchi K, et al. Neurodevelopmental and personality traits of somatic symptom disorder: a cross-sectional study. *PCN Rep.* 2025;4(1):e70082. doi:10.1002/pcn5.70082
- Hallberg H, Maroti D, Lumley MA, et al. Internet-delivered emotional awareness and expression therapy for somatic symptom disorder: one year follow-up. *Front Psychiatry*. 2024;15:1505318. doi:10.3389/fpsyt.2024.1505318
- 3. Jongsma K, Darboh BS, Davis S, et al. A cognitive behavioural group treatment for somatic symptom disorder: a pilot study. *BMC Psychiatry*. 2023;23(1):896. doi:10.1186/s12888-023-05141-9
- Tuttle MC, Ciampa DJ, Godena E, et al. The evaluation and treatment of somatic symptom disorder in primary care practices. Prim Care Companion CNS Disord. 2024;26(1). doi:10.4088/PCC.23f03549.
- 5. Hyun Seo E, Kim SG, Yoon HJ. Risperidone augmentation in antidepressant-resistant somatic symptom disorder. *Psychiatry Clin Psychopharmacol.* 2025;35(1):88–91. doi:10.5152/pcp.2025.24931
- Boucherie DE, Reneman L, Ruhé HG, et al. Neurometabolite changes in response to antidepressant medication: a systematic review of (1)H-MRS findings. *Neuroimage Clin.* 2023;40:103517. doi:10.1016/j.nicl.2023.103517
- 7. Nagoshi Y, Tominaga T, Fukui K. Blonanserin augmentation for treatment-resistant somatic symptom disorder: a case series. *Clin Neuropharmacol.* 2016;39(2):112–114. doi:10.1097/WNF.00000000000134
- 8. Calsolaro V, Femminella GD, Rogani S, et al. Behavioral and psychological symptoms in dementia (BPSD) and the use of antipsychotics. *Pharmaceuticals*. 2021;14(3):246. doi:10.3390/ph14030246
- 9. Awuah WA, Kalmanovich J, Mehta A, et al. Multilevel pharmacological effects of antipsychotics in potential glioblastoma treatment. *Curr Top Med Chem.* 2023;23(5):389–402. doi:10.2174/1568026623666230102095836
- 10. Rognoni C, Bertolani A, Jommi C. Second-generation antipsychotic drugs for patients with schizophrenia: systematic literature review and meta-analysis of metabolic and cardiovascular side effects. *Clin Drug Investig*. 2021;41(4):303–319. doi:10.1007/s40261-021-01000-1
- 11. Li R, Fu R, Cui Z-Q, et al. Effects of low-frequency rTMS combined with risperidone on the gut microbiome in hospitalized patients with chronic schizophrenia. *Brain Res.* 2023;1819:148539. doi:10.1016/j.brainres.2023.148539
- 12. Sheng R, Chen C, Chen H, et al. Repetitive transcranial magnetic stimulation for stroke rehabilitation: insights into the molecular and cellular mechanisms of neuroinflammation. Front Immunol. 2023;14:1197422. doi:10.3389/fimmu.2023.1197422
- 13. Wang M, Lu S, Hao L, et al. Placebo effects of repetitive transcranial magnetic stimulation on negative symptoms and cognition in patients with schizophrenia spectrum disorders: a systematic review and meta-analysis. *Front Psychiatry*. 2024;15:1377257. doi:10.3389/fpsyt.2024.1377257
- 14. Li X, Yuan X, Kang Y, et al. A synergistic effect between family intervention and rTMS improves cognitive and negative symptoms in schizophrenia: a randomized controlled trial. J Psychiatr Res. 2020;126:81–91. doi:10.1016/j.jpsychires.2020.04.009
- 15. Huang J, Zhong Y, Duan Y, et al. Case report: new insights into persistent chronic pelvic pain syndrome with comorbid somatic symptom disorder. *Front Psychiatry*. 2023;14:1119938. doi:10.3389/fpsyt.2023.1119938
- 16. Klein MO, Battagello DS, Cardoso AR, et al. Dopamine: functions, signaling, and association with neurological diseases. *Cell Mol Neurobiol*. 2019;39(1):31–59. doi:10.1007/s10571-018-0632-3
- 17. Zarrindast MR, Khakpai F. The modulatory role of dopamine in anxiety-like behavior. Arch Iran Med. 2015;18(9):591-603.
- 18. Delli Pizzi S, Franciotti R, Ferretti A, et al. High γ-Aminobutyric acid content within the medial prefrontal cortex is a functional signature of somatic symptoms disorder in patients with parkinson's disease. *Mov Disord*. 2020;35(12):2184–2192. doi:10.1002/mds.28221
- 19. Park B, Lee S, Jang Y, et al. Affective dysfunction mediates the link between neuroimmune markers and the default mode network functional connectivity, and the somatic symptoms in somatic symptom disorder. *Brain Behav Immun.* 2024;118:90–100. doi:10.1016/j.bbi.2024.02.017
- Yapa P, Haque MS. Use of the international classification of diseases-10, (ICD 10) to recognise pervasively hyperactive children in a child-guidance clinic: feasibility and validity. *Eur Arch Psychiatry Clin Neurosci.* 1991;240(3):138–143. doi:10.1007/BF02190753
- Deligiannidis KM, Meltzer-Brody S, Gunduz-Bruce H, et al. Effect of zuranolone vs placebo in postpartum depression: a randomized clinical trial. JAMA Psychiatry. 2021;78(9):951–959. doi:10.1001/jamapsychiatry.2021.1559
- 22. Meng J, Du J, Diao X, et al. Effects of an evidence-based nursing intervention on prevention of anxiety and depression in the postpartum period. *Stress Health.* 2022;38(3):435–442. doi:10.1002/smi.3104
- 23. Sansom GT, Kirsch K, Horney JA. Using the 12-item short form health survey (SF-12) to assess self rated health of an engaged population impacted by hurricane Harvey, Houston, TX. *BMC Public Health*. 2020;20(1):257. doi:10.1186/s12889-020-8349-x
- 24. Tabatabaei MS, Ahmed M. Enzyme-linked immunosorbent assay (ELISA). Methods Mol Biol. 2022;2508:115-134.
- 25. Dalhuisen I, van Oostrom I, Spijker J, et al. rTMS as a next step in antidepressant nonresponders: a randomized comparison with current antidepressant treatment approaches. *Am J Psychiatry*. 2024;181(9):806–814. doi:10.1176/appi.ajp.20230556
- 26. Ren Z, Mu L, Wang L, et al. Predictive role of impulsivity, anxiety, and depression in the efficacy of intermittent theta burst transcranial magnetic stimulation modalities for treating methamphetamine use disorder: a randomized clinical trial. J Subst Use Addict Treat. 2024;156:209189. doi:10.1016/j.josat.2023.209189

- Yin ZL, Ge S, Huang L-H, et al. [Acupuncture combined with repetitive transcranial magnetic stimulation for post-stroke depression: a randomized controlled trial]. *Zhongguo Zhen Jiu*. 2022;42(11):1216–1220. doi:10.13703/j.0255-2930.20211221-0002
- Chen QM, Yao F-R, Sun H-W, et al. Combining inhibitory and facilitatory repetitive transcranial magnetic stimulation (rTMS) treatment improves motor function by modulating GABA in acute ischemic stroke patients. *Restor Neurol Neurosci*. 2021;39(6):419–434. doi:10.3233/RNN-211195
- 29. Pan B, Lian J, Deng C. Chronic antipsychotic treatment differentially modulates protein kinase A- and glycogen synthase kinase 3 beta-dependent signaling pathways, N-methyl-D-aspartate receptor and gamma-aminobutyric acid A receptors in nucleus accumbens of juvenile rats. J Psychopharmacol. 2018;32(11):1252–1263. doi:10.1177/0269881118788822
- 30. Gener T, Tauste Campo A, Alemany-González M, et al. Serotonin 5-HT(1A), 5-HT(2A) and dopamine D(2) receptors strongly influence prefronto-hippocampal neural networks in alert mice: contribution to the actions of risperidone. *Neuropharmacology*. 2019;158:107743. doi:10.1016/j.neuropharm.2019.107743
- Yang L, Su Y, Guo F, et al. Deep rTMS mitigates behavioral and neuropathologic anomalies in cuprizone-exposed mice through reducing microglial proinflammatory cytokines. *Front Integr Neurosci*. 2020;14:556839. doi:10.3389/fnint.2020.556839

Neuropsychiatric Disease and Treatment



Publish your work in this journal

Neuropsychiatric Disease and Treatment is an international, peer-reviewed journal of clinical therapeutics and pharmacology focusing on concise rapid reporting of clinical or pre-clinical studies on a range of neuropsychiatric and neurological disorders. This journal is indexed on PubMed Central, the 'PsycINFO' database and CAS, and is the official journal of The International Neuropsychiatric Association (INA). The manuscript management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/neuropsychiatric-disease-and-treatment-journal

